\documentstyle[12pt,twoside,macpsfig]{article}
\input{macros-course}

\begin{document}

\handout{ps1}{HANDOUT TITLE}{DATE OF HANDOUT, 1994}

\pset{1}

{\bf Reading:\ }Maurer \& Ralston: Section 0.7, Chapter 2, Sections 7.1--3,
and Sections 7.5-7.6.

\medskip

The problems are divided into oral problems (prefixed~``{\bf O}'') and
written problems (prefixed~``{\bf W}'').
The oral problems are due in tutorial on Monday, February 14.  
The written problems are due at the beginning of class on [[[DATE]]].
Remember to write up each problem on a separate page or
pages, with your name on each sheet.  
Read Handout~\ref{course-information} for the policy on problem sets.

\bigskip

\begin{problems}


\oproblem This is oral problem number one...
...
\wproblem This is written problem number seven...
...

\end{problems}


\end{document}


